Author's response to reviews

Title: Enhanced surveillance of invasive listeriosis in the Lombardy region, Italy, in the years 2006-2010 reveals major clones and an increase in serotype 1/2a

Authors:

Caterina Mammina (caterina.mammina@unipa.it)
Antonio Parisi (parisi.izs@tin.it)
Anna Guaita (anna.guaita@unimi.it)
Aurora Aleo (auroraaleo@libero.it)
Celestino Bonura (celestino.bonura@unupa.it)
Antonino Nastasi (antonino.nastasi@unifi.it)
Mirella Pontello (mirella.pontello@unimi.it)

Version: 4 Date: 23 February 2013

Author's response to reviews: see over
Dear Editor,

I am submitting the revised version of the manuscript by Mammina C et al. entitled “Enhanced surveillance of invasive listeriosis in the Lombardy region, Italy, in the years 2006-2010 reveals major clones and an increase in serotype 1/2a”.

I am attaching below the answers to the comments of reviewers.

I hope the manuscript in the present version could be accepted for publication on BMC Infectious Diseases.

Sincerely

Caterina Mammina

Caterina Mammina, MD
Department. of Sciences for Health Promotion "G. D'Alessandro"
Via del Vespro 133
I-90127 Palermo, Italy
phone +39 0916553623
FAX +39 0916553641
email caterina.mammina@unipa.it